Author’s response to reviews

Title: The improvement of QRS-T angle as a manifestation of reverse electrical remodeling following renal transplantation in end-stage kidney disease patients on haemodialysis

Authors:

A Jaroszynski (jaroszynskiaj@interia.pl)
Jacek Furmaga (jacekf61@o2.pl)
Tomasz Zapolski (zapolia@wp.pl)
Tomasz Zaborowski (tzaborowski@wp.pl)
Sławomir Rudzki (slawomir.rudzki@umlub.pl)
Wojciech Dąbrowski (w.dabrowski5@gmail.com)

Version: 2 Date: 05 Nov 2019

Author’s response to reviews:

Dear Editor,

Thank you very much once again for taking your time to review our paper “The improvement of QRS-T angle as a manifestation of reverse electrical remodeling following renal transplantation in end-stage kidney disease patients”.

I did my best to address all concerns in the paper as well as in responses to their particular questions, which are enclosed with the manuscript.

Here are my responses to the comments:

1. We have tried to reformulate some section of text to eliminate significant overlaps;

2. We have moved the keywords to the end of abstract;
3. In the availability of data and materials section we have stated who should be contacted if someone wants to request the data;

4. We have described the role of the funding body in the funding section;

5. We have corrected the number of table 5 and 6;

best wishes,

Andrzej Jaroszyński